Journal of Knowledge Management



Corporate Social Responsibility as a Catalyst of Circular Economy? A Case Study Perspective in Agri-food

Journal:	Journal of Knowledge Management
Manuscript ID	JKM-06-2022-0451
Manuscript Type:	Research Paper
Keywords:	Corporate Social Responsibility, Circular Economy, Stakeholder Theory, Agri-food, Environmental Innovation

SCHOLARONE[™] Manuscripts

Corporate Social Responsibility as a Catalyst of Circular Economy? A Case Study Perspective in Agri-food

Abstract

Purpose – This research aims to analyze how, under the stakeholder theory, Corporate Social Responsibility (CSR) might favor the emergence of Circular Economy (CE) in the Agri-food sector, which is a relevant context as it is technologically dynamic and requires paying attention to all the stakeholders.

Design/methodology/approach – It has been adopted an exploratory, qualitative research design to study the phenomenon in detail as it facilitates the understanding of complex phenomena such those under investigation, and helps enrich existing theory with new insights from real-world cases to add theoretical generalizations to the existing body of research in the field.

Findings – The results of the study highlight that companies adopting CSR models are oriented towards circularity.

Practical implications – This research provides useful indications to managers and policy makers as to how to favor the two approaches and benefit all the stakeholders.

Originality/value – While there is wide scholarly and managerial interest towards CSR and CE, previous research has mainly analyzed CE and CSR as two independent phenomena. Therefore, there is a lack of understanding about how the two areas are linked. Following previous studies that have started to theoretically argue an interconnection between CSR and CE, in this research it has been empirically investigate, and further explore theoretically, whether CSR can implicitly encourage the emergence of CE approaches.

<text>

1. Introduction

The Ellen MacArthur Foundation, a leading organization aiming to spread the adoption of the Circular Economy (CE), defines the CE as "an umbrella term for an economy designed to regenerate itself in which waste is minimized" (Ellen-MacArthur Foundation, 2013). This approach requires a paradigm shift that can prioritize reducing waste and pollution, favour resource efficiency and encourage repairs rather than replacement mechanisms (Kirchherr *et al.*, 2017; Cainelli *et al.*, 2020).

The CE can help attain the Sustainable Development Goals (SDGs) set by the United Nations. Indeed, CE can be traced back to the SDG 12 "Responsible Consumption and Production," one of the 17 SDGs included in the 2030 Agenda, the action program for the planet adopted in September 2015 by the United Nations. SDG 12 seeks to ensure sustainable production and consumption models for a better world. Thus, CE may aid in tackling grand challenges, i.e., pressing environmental and social problems that afflict society (Berrone et al., 2013; Cappa et al., 2020; Centobelli et al., 2020; Sakshi et al., 2020), by minimizing waste and increasing reuse and recycle of materials benefiting the environment and society on the one hand (Pomponi and Moncaster, 2017; Fehrer and Wieland, 2021) and by improving operations and production benefiting the economic performance on the other (Parida et al., 2019; Ranta et al., 2018a). The objective is to innovate the relationship between the organizations and the surrounding environment by implementing a closed loop of regeneration and restoration (Fernandez de Arroyabe et al., 2021; Barreiro-Gen and Lozano, 2020). The CE model is also a lever to create new jobs for developing efficient systems able to postpone the end of life of the products themselves (Heves et al., 2018; Patwa et al., 2021). Therefore, CE aims to nurture sustainable development satisfying the needs of all the stakeholders involved (Rainville, 2021; Burger et al., 2019; Marrucci et al., 2021). In this respect, knowledge management, i.e. the effective management of intellectual resources that support the creation, transfer and application of knowledge within organizations, is becoming crucial for

understanding CE and spread its diffusion (Grover and Davenport, 2001; Ghisellini *et al.*, 2016). Indeed, a close knowledge-related collaboration with all the stakeholders can ease the emergence of CE (Zucchella and Previtali, 2019; Govindan and Hasanagic, 2018; Zhang *et al.*, 2021; Gomes *et al.*, 2021; Vendrell-Herrero, 2021). In addition to what done and known so far, there is a ferment around what can be a further catalyst of CE (Seles *et al.*, 2022; Ciliberto *et al.*, 2021).

Businesses have increasingly been committing to another sustainability-oriented managerial strategy, i.e. Corporate Social Responsibility (CSR). CSR is the strategic orientation of an organization to implement socially and environmentally responsible actions while still pursuing economic goals (Franco et al., 2020; Russo and Perrini, 2010; Ghasemzadeh et al., 2021; Ltifi and Hichri, 2022), aiming to create value for all the stakeholders (Radu and Smaili, 2021). Indeed, customers, suppliers, employees and policymakers increasingly demand firms adapt their business strategies to jointly enhance social, environmental and economic performance (Cassar and Meier, 2018; Maon et al., 2021). Also CSR is a knowledge intensive phenomenon, and companies are interested in how to accumulate, create and share knowledge to effectively implement CSR throughout the organization (Gangi et al., 2019). While there is wide scholarly and managerial interest towards CSR and CE, previous research has mainly analyzed CE (Blomsma, 2018; Geissdoerfer et al., 2017; Ghisellini et al., 2016; Korhonen et al., 2018; Lewandowski, 2016; MacArthur, 2013; Manninen et al., 2018; Morseletto, 2020; Ranta et al., 2018b; Saidani et al. 2019; Tukker 2015) and CSR (Barrena-Martínez et al. 2015; Gangi et al. 2018; Perrini et al., 2007; Santos, 2011; Vázquez-Burguete et al., 2017) as two independent phenomena. Therefore, there is a lack of understanding about how the two areas are linked. Since both are aimed at enhancing all three pillars of sustainability, i.e., economic, social and environmental (Cappa et al., 2020; Hansmann et al., 2012), we argue that they can be mutually helpful.

In greater detail, following previous studies that have started to theoretically argue an interconnection between CSR and CE (Daú *et al.*, 2019; Esken *et al.*, 2018; Leandro and Paixao, 2018; Velenturf *et al.*, 2019), in this research we empirically investigate, and further explore theoretically, whether CSR can implicitly encourage the emergence of CE approaches. In particular, the research question we addressed in this research is the following: *How does CSR favour the emergence of CE?* To answer this query, in this study we have analysed eight case studies in the Agri-food sector, which is a relevant context where to conduct our research as it is technologically dynamic and requires paying attention to the stakeholders. Therefore, it represents a fertile ground where to study the joint application of CSR, which has been evidenced to be increasingly relevant in food related matters (Kong *et al.*, 2019; Kong, 2012), and CE, which is emerging due to the growing technological advancements (Pieroni *et al.*, 2019; Berrone *et al.*, 2013; Cainelli *et al.*, 2020) and is attracting growing attention in the food sector (Halloran *et al.*, 2014; Mylona *et al.*, 2018; De Schutter *et al.*, 2020).

The paper has implications for theory, practice and policymaking. First, it enriches the scientific knowledge of CSR and CE phenomena, and contributes to the understanding of their interconnections. Second, it underlines that stakeholder theory (Freeman and Evan, 1990) is a theoretical lens that can be effectively used to understand how CSR and CE are correlated. Moreover, we have empirically shown how CSR can favour the transition towards CE, although in an unstructured way. In addition, this research also informs managers about the fact that if they undertake certain CSR activities, they are also moving towards CE. More broadly, the outcomes are also of interest to policy makers, who can promote the emergence of CSR by evidencing that it also favors the establishment of CE. In such a way, it possible to push thus companies towards the joint implementation of CSR and CE, benefiting all the stakeholders.

The manuscript is structured as follows: Section 2 reviews the literature on CE and CSR. Section 3 discusses the relevance of the Agri-food sector and describes the methodology adopted and the structure of the data collection phase. Section 4 provides findings. Section 5 discusses results and implications for theory, practice and policymaking, while Section 6 concludes the analysis of the multiple case studies, highlighting the limitations of the work and future research directions.

2. Background

In the 1970s a new model that was in opposition to classical linear systems and that was suitable for the improvement of society and human wellbeing started being developed. Song (1972) systematized parameters into a mathematical model that could give life to a new science that would take into account the scarcity of resources and the population and the consequent environmental degradation. However, it is with the birth of the Ellen MacArthur Foundation in 1976 that the term CE becomes concrete. The Foundation aims to promote the various principles of CE by incorporating several schools of thought about it into a single integrated concept (Ellen MacArthur Foundation, 2015, Linstead *et al.*, 2014). In 2020 the European Commission (EU) presented a package regarding CE called "The Missing Link: A European Action Plan for the CE" that considers CE as an opportunity for growth for all stakeholders and for the implementation of specific objectives based on the incentive policies of this model (European Commission, 2015; FEEM, 2020; WHO - Regional Committee for Europe, 2014).

As governments and businesses increase their attention towards environmental innovations (Dias Angelo *et al.*, 2012; Marrucci *et al.*, 2021), i.e., innovative solutions able to reduce intensive consumption of resources as well as sustain economic growth (Geissdoerfer *et al.*, 2018; Ghisellini *et al.*, 2016), CE is becoming central as it is able to bring economic, social and environmental benefits (Suchek *et al.*, 2021; Barreiro-Gen and Lozano, 2020). It has been

argued that the turning point of change is to be found in design and not in economic activity itself, based on the growth of "cradle to cradle" (Lawrence, 2013), thus abandoning the old "cradle to grave" system (Song et al., 2018). Indeed, CE may aid in addressing grand challenges and benefit sustainable development, by creating competitive advantage through innovative and sustainable business models (Kristoffersen et al., 2020; Fernandez de Arroyabe et al., 2021; Ferasso et al., 2021). The advantages of CE include the promotion of growth by ensuring competitiveness through opportunities for social cohesion and integration as well as paying attention to the environment (Loiseau et al., 2016; Pan et al., 2018; Postolache and Troaca, 2018; Urbinati et al., 2017). The transition towards CE affects and involves all the actors of society, from producers to institutions, down to the consumer (Ciulli and Kolk, 2019; D'Amato et al., 2017; Geissinger et al., 2019; Hanley and Semrau, 2022). As can be gleaned from the above-mentioned benefits, the CE model can be analyzed through the "stakeholder theory" perspective (Govindan and Hasanagic, 2018), because all the stakeholders, e.g. government, communities, trade unions, consumers, employees, suppliers, and citizens, impact CE and are also affected by it (Hussainey and Salama, 2010). Another stakeholders-oriented approach adopted by companies is CSR (Salvioni and Almici, 2020). Already in the 1960s, Frederick (1960) had underlined that it was necessary to reallocate managers with responsibility to focus their attention on society as well that of their company. In the same decade, McGuire (1969) highlighted how companies have, in addition to obligations in the social and economic spheres, extended responsibilities that go well beyond those dictated by regulatory obligations, which require incisive actions and direction on social policies. Friedman (2017) emphasized how companies were mainly focused on maximizing profits but instead there is also need also to benefit shareholders. CSR spawned from these considerations, and the most comprehensive definition has been provided by Carroll (1991): "Corporate social responsibility encompasses the economic, legal, ethical,

and discretionary (philanthropic) expectations that society has of organizations at a given point in time". The interest towards CSR has continued to grow through the years. The European Commission (2001), in the "Green Paper Promoting a European Framework for Corporate Social Responsibility", highlighted that CSR draws attention to the environment and social issues through the activities companies carry out, reaching all the stakeholders involved. Werther and Chandler (2005) discussed the importance of CSR as a vehicle to enhance the corporate brand. Porter and Kramer (2006) affirmed how CSR can be a means to adapt their corporate strategies to the needs of the society. Indeed, also Heslin and Ochoa (2008) underlined how CSR, in addition to creating of social value, can positively affect all the stakeholders involved. Moreover, CSR fosters relationships between a company and its employees through specific initiatives aimed at their safety and training, including targeted welfare and benefit policies (Steurer, 2010). From an environmental point of view, many benefits range from low environmental impact production, to the adoption of certifications, to supplier traceability for greater transparency and traceability as well as innovative plant technologies (Baumgartner, 2014; Lucchini and Moisello, 2017). From a financial perspective, companies with high levels of CSR have better performance (Franco et al., 2020; Gangi et al., 2019). Therefore, CSR allows companies to pursue activities that bring about simultaneously social, environmental and economic benefits (Cappa et al., 2020; Lubin and Esty, 2012; Russo and Perrini, 2010; Ltifi and Hichri, 2022). For these reasons, also in the case of CSR previous research has extensively grounded their study on stakeholder theory (Dmytriyev et al., 2021; Franco et al., 2020; Waheed and Zhang, 2020; Freeman and Dmytriyev, 2017; Theodoulidis et al., 2017; Cordeiro and Tewari, 2015; Russo and Perrini, 2010; Radu and Smaili, 2021).

Considering the common focus on stakeholders and sustainable development objectives, we argue that CSR and CE approaches could be mutually beneficial. We contend that CSR,

which is a broader and more widespread self-regulatory model, can favour the emergence of principles and models of the recent CE phenomenon. While such relation has been theorized by few previous studies (Daú *et al.*, 2019; Esken *et al.*, 2018; Leandro and Paixao, 2018; Velenturf *et al.*, 2019), we contribute to this gap by deepening its theoretical understanding and by empirically highlighting how this happens, to thus enlighten scholars, managers and policymakers towards a more widespread implementation of CE. In particular, we do so with a qualitative approach in the Agri-food sector, which constitutes a relevant case study, as detailed in the following section.

3. Data and methods

3.1 Research context: The Agri-food sector

The Agri-food - also referred to as agribusiness - is the second largest manufacturing industry (Bolzani *et al.*, 2015), and has undergone numerous changes recently to better satisfy consumers and all the stakeholders involved in light of the rapid technological innovations and demands for sustainability (European Commission, 2019; Del Vecchio *et al.*, 2022; Fait *et al.*, 2019). As a result, this market is evolving towards a perspective that aims more at quality than quantity (Nasir and Karakaya, 2014), with comprise increasing attention to health aspects of food (Román *et al.*, 2017, Ciravegna and Brenes, 2016). Therefore, companies must be able to solve pressing problems concerning consumers and stakeholders, i.e., they must offer healthy and quality products by producing and delivering goods and services in a sustainable way as well as communicating and interacting with the stakeholders so as to be increasingly important in the food sector due to the wide range of benefits (Kong, 2012; Kong et al., 2019).

Today's consumers are increasingly attentive to their health and more aware of the purchases

they make. The policy adopted in the Agri-food market to date is aimed at focusing greater attention both on food-product quality and, at the same time, on environmental protection (Spalding et al., 2014). This dual focus is essential also for companies to guide their business decisions. From this perspective, technological innovation is useful to adapt to market demand and thus pay attention to quality and environmental aspects, while keeping prices affordable for the consumer (Kamilaris et al., 2019). As a consequence, the impact of digital transformation on the Agri-food sector and on the ability to adequately respond to sustainability requirements is increasingly important (Liu et al., 2011; Colbert et al., 2016). New technologies make it easier to understand availability, prices and performance of a product or service (Modgil et al., 2021; Venkatraman, 2017; Warner, 2019). In fact, in recent years, digital technology capabilities have become one of the strategic assets of Agri-food as companies modernize various aspects of their production (Annosi et al., 2019). The innovations include the technologies that monitor equipment and management software, the correct management of production processes (a distinguishing factor for product quality), traceability and food safety (the basis of market competitiveness), optimal management of crops (sowing and harvesting) (Appio et al., 2021). Digitization, even after the pandemic, has made it possible to limit environmental damage to favour more sustainable working methods. In fact, even very distant companies have the opportunity to collaborate with one another and exchange information, attaining enormous economic advantages and with reduced environmental impact. Improvements can be made regarding citizens' health, education and living conditions; decarbonisation and the increase of renewable and clean energy; and responsible models in the fields of transport, construction and the Agri-food sector (Modgil et al., 2021).

In addition, the supply of Agri-food products may benefit by advancements that allow fast product delivery to permit launching new products more frequently (Kikuchi and Kanematsu,

2019). Another important factor in the current panorama of the technologically dynamic food supply chain is big data (Ardito *et al.*, 2019; Del Vecchio *et al.*, 2018; Elia *et al.*, 2019; Acciarini *et al.*, 2020). In fact, big data management can ensure the customer a high degree of food safety and traceability as well as product conformity to established standards (Aung and Chang, 2014). Technology aids the consumer not only by providing product guarantees but also by communicating transparency of operations (Kane *et al.*, 2015). Another IT advancement is block chain, which allows storing information safely and quickly to favour dialogue between all interested stakeholders (Kamath, 2018). Through the simplification of information exchange, it is possible to protect consumers from counterfeiting, which is an increasing phenomenon (Brewster *et al.*, 2017), for example using intelligent labels that track food products in all food sectors throughout their journey, to guarantee quality and safety to final consumers (Surasak *et al.*, 2019).Thanks to these recent technological advancements, Agri-food is reducing its environmental footprint and improving the product quality to satisfy customers and stakeholders (Sharma *et al.*, 2018; Lernon and Verhoef, 2016).

Given these considerations, Agri-foodis a relevant context in which to conduct our study, as it is a dynamic environment both due to the pressing attention of its stakeholders to the quality and sustainability of the outcomes (Shepherd *et al.*, 2018)., as well as to the several technological advancements that are undergoing (Tian, 2016; Brenes *et al.*, 2020). As a consequence, it represents an interesting field where to explore the joint application of CSR and CE, as the former has been started being considered in food related matters (Kong *et al.*, 2019; Kong, 2012), and the latter is growing thanks to technological advancements (Pieroni *et al.*, 2019; Berrone *et al.*, 2013; Cainelli *et al.*, 2020).

3.2 Research methodology

In answering the research question of our study, we have adopted an exploratory, qualitative research design to study the phenomenon in detail as it facilitates the understanding of

complex phenomena such those under investigation (Fleming, 2001; Yin, 2009), and helps enrich existing theory with new insights from real-world cases to add theoretical generalizations to the existing body of research in the field. This method allowed us to explore in depth and provide substantial support for the development of preliminary theoretical development concerning CE and CSR (Flick *et al.*, 2004; Siggelkow, 2007). To this end, we considered eight relevant case studies that allowed us to reach theoretical saturation. We aim at enriching the literature in this field with new insights from real-world cases in the context of Agri-food. In particular, we have analysed agri-business companies operating in Italy, a reference country when dealing with food related aspects, in search of similarities and differences to finally add theoretical generalizations to the existing body of research concerning how CSR processes can favour the emergence of CE in this sector.

In setting up the multiple case study (Yin, 2009), we established a sampling framework of criteria associated with the theoretical background and research interest of our study: i) the case firms had to be organizations well known for the high quality and value of their products, thus constituting the perception with which the consumer bases his choices for purchasing; ii) the case firms had to have a solid reputation for innovation, quality of management, personnel and customer management, financial stability, social stability and exclusive processing of production techniques, thus constituting added value to the products, making them more palatable in the market; and iii) the case firms had to have adopted CSR approaches, inferable from related documents publicly available, i.e., sustainability reports.

We focused on Italian Small and Medium-sized Enterprises (SMEs) active in the Agri-food sector, as SMEs are the most numerous typologies of firms active in this sector, and we were able to identify eight organizations meeting the above-mentioned criteria for which information is available (i.e., Organization F; Organization GC; Organization DMA; Organization S; Organization VDOSMTA; Organization VC; Organization L; Organization

GM, by the initial letters of each organization). In fact, Italy is a relevant context for studying the Agri-food sector because the stakeholders active in the country are extremely careful regarding quality and sustainability aspects. Globally, Italy is the sixth largest exporter in the Agri-food sector. It ranks second among European producers in the percentage of companies in the Agri-food that have introduced product and process innovations. Moreover, it is also technologically advanced compared to other countries based on investments made in Information and Communication Technologies (ICT). Finally, among the sectors of the national economy, Agri-food is the most resilient (Food and Agriculture Organization of the United Nations, 2020). We focused on Italian SMEs as they constitute 75% of the total number of enterprises in Italy and thus are the backbone of the national production system, with 80% of the workforce. SMEs contribute to increasing the entrepreneurial level as well as innovation, and therefore become a means for promoting the competitiveness and development of the territory (European Commission, 2021). For confidentiality reasons, their names are disguised. Appendix 1 compares the main characteristics of the sampled firms. To evaluate how the CSR orientation of the eight organizations lead to the adoption of CE practices, the sustainability reports - which indicate efforts towards CSR - of each company were analysed. We have evidenced the parts of the report where CE principles were clearly evident, referring to, e.g., the 9R model of CE (Kirchherr et al., 2017), which interprets the circular orientation of the single organizations considered in our research. This model identifies the nine strategies (R0 - Refuse, R1 - Rethink, R2 - Reduce, R3 - Reuse, R4 -Repair, R5 - Refurbish, R6 - Remanufacture, R7 - Repurpose, R8 - Recycle, R9 - Recover)

that characterize the gradual transition from linear economy to CE (Figure 1). We checked the adoption and level of increase in the aforementioned strategies within the sustainability report of each single organization, to verify their orientation towards circularity.

--- Insert Figure 1 about here ---

4. Results

We have analysed the sustainability reports that have been made publicly available in 2020 by the SMEs being considered that were active in the Agri-food sector in Italy. These reports constitute a tool that takes into consideration the economic, social and environmental impacts of an organization's activity as well as the expectations of stakeholders, allowing for the achievement of strategic and business objectives. The presence of sustainability reports highlights that the companies are active in the field of CSR. We have analysed the presence of the 9Rof CE identified by (Kirchherr *et al.*, 2017), as reported in Figure 1, in these sustainability reports. The results of such analysis, conducted independently by four scholars active in the field of finance and innovation, led to the following results, as also summarized in Figure 2.

Analysis of the sustainability report of Organization F shows that the "company is historically attentive to the quality control of products with an accurate choice of raw materials" and "it has been operating for years in favour of the environment through actions to reduce paper, separate paper collection, recycle plant water, reduce waste, regenerate used cartridges, use anti-pollution paints and care for the greenery around the company itself". It is thus clear that Organization F has completely adopted the new way of managing its business (Newell 2015) in line with R1 (rethinking), which highlights the creation of a business model and shared values, thereby rethinking the way of doing business. It also pays particular attention to reducing (R2) paper, cartridges, batteries, oils and packaging. Waste management is an important strategy for the Organization, which can activate procedures and systems aimed at improving the entire process of waste - from its production to its disposal - and which involves various lighthouses up to the reuse of waste materials. It also allows for positive effects on human health and the environment, thus saving and recovering natural resources and optimizing their management. Waste in landfills is significantly reduced and the weight and volume of containers also decreases. In line with the principle of R6 (remanufacture) and

R8 (recycle), the company has decided to use recycled paper and remanufactured cartridges with significant economic savings, and thus significantly reduce pollution. The production of recycled paper in fact consumes less water and energy than the production derived from natural resources, reducing the number of waste and CO2 emissions in the atmosphere in this context the company responds to the principle of "reuse" (R3) and uses photo catalytic, anti-pollution and antibacterial paints.

Organization GC, in its sustainability report, highlights that "*it has rethought its own reorganization vis-à-vis the environment and reduced the costs related to its use of water resources by about 40% as well as those related to lighting by using energy-saving lamps, which have cut consumption by 30%*". Moreover, "*the company recovers sludge, which is then destined for compost; production waste is transferred to other companies and the remaining materials are disposed of and differentiated*". So, Organization GC has adapted its entrepreneurial activity to a new way of thinking and redesigned products in new business processes with a view to efficiency, which also directs production in qualitative and quantitative performance, thus responding to the "rethink" (R1) principle. The company reduces the costs of using water resources and the cost for lighting in line with the "reduce" (R2) and "recycle" (R8) principles. This brings to improving energy efficiency and thus improving performance: e.g., energy consumption to increase energy efficiency or to exploit energy in the best possible way could consist of simple actions such as replacing the lighting system with an energy-saving one or using a wattmeter to monitor consumption (Corsini et al., 2019).

The sustainability report of Organization DMA states that "the company has replaced its own cars with hybrid cars also with a view to rethinking the company's policy and sustainable processes, thus allowing it to prevent and reduce pollution as well as improve company performance". In addition, "it has created an electrical and thermal energy system able to

almost entirely satisfy the company's needs; replaced plastic with eco-compatible materials; and reduced the packaging of confections, waste food, and use of waste products in other processes". In so doing, Organization DMA has rethought new ways of moving the company outside the schemes of the linear economy into a circular supply chain perspective (Pellegrini *et al.*, 2020), responding to the "rethink" (R1) principle and to an engineering process that requires sharing between specialized figures and all the stakeholders involved. In compliance with the "reduce" (R2) and "reuse" (R3) principles, it reduces energy consumption and the reuse of water and waste production, CO2 emissions and plastic packaging to minimize the negative effects of anthropogenic activity on climate change. Paper and plastic are the main waste produced and the actions taken are aimed at reducing the paper and packaging consumption as well as increasing the recycling (R8) of these products. To reduce CO2 emissions into the atmosphere, Organization DMA encourages car sharing and the use of public transport for its employees. Finally, it uses discarded products or parts of them in new products with different functions (R7).

In Organization S's sustainability report, "the company proposes responsible management of the wastewater from milk processing and the recovery of resources, including large quantities of water thanks to the use of whey concentrator" and "highlights contained energy consumption thanks to renewable energy sources, which are implemented by a co generator for energy production". Also "through the company's Green Strategy, large quantities of drinking water are saved; the water obtained from whey is recycled for industrial washing and the internal production of electricity occurs with photovoltaic panels". This means that Organization S responds to the rethink (R1) strategy by incorporating the principle of the sharing economy through a model that aims to share and optimize consumption for the redistribution of goods and services and by instituting virtuous behaviour. Moreover, it responds to the "reduce" (R2) principle by creating a by-product, namely whey, from

mozzarella-production waste. The by-product route is important, because through it a company can implement a virtuous path and enter into the perspective of CE. From this point of view, it can implement corporate-saving processes by configuring the activity as new corporate business, thus allowing it to also reduce corporate costs while operating in the recycling (R8) and remanufacturing (R6) perspectives. After concentrating the whey, it can be sold as a secondary raw material to food companies, thereby repurposing them(R7),or reusing internally in keeping with the "reuse" (R3) principle. Furthermore, regarding the reduce principle, the organization reduces its water and energy consumption. These processes lead to saving resources and better using waste while reducing the environmental impact.

Organisation VDOSMTA states, in its sustainability report, that "choosing raw materials, recyclable containers, energy recovery processes and environmentally conscious suppliers allows it to produce sustainably for the community and future generations. [...] The plant's waste water is monitored and continuously kept under control, to prevent possible deviations from the legal standards. The waste is collected separately and disposed of by companies authorised to treat the various materials. [...] Atmospheric emissions are kept under control, as is the impact of the noise produced by its operations on its neighbours". Thus, Organisation VDOSMTA responds to the rethink (R1) strategy, implementing business models that are the basis of CE with sharing platforms that put product owners in contact with organizations or individuals, thereby increasing productivity for shared access. Its activity has an environmental impact that respects the community and the environment in which it is located, through its choice of raw materials, recyclable (R8) containers, energy recovery processes, and environmentally friendly suppliers, in compliance with the "reduce" (R2) principle. Its waste is collected separately and is disposed of by companies authorised to treat the various materials; it reduces its use of plastic packaging by effectively pursuing its policy of using as little as necessary. Moreover, the company favours recyclable or compostable

packaging materials. It reuses (R3) packaging whenever possible, even several times, by reusing energy within business processes. Its system drainage is constantly monitored and controlled, to prevent possible deviations from the legal standards and its system's water is reused with a view to recycling. Atmospheric emissions are kept under control, as is the impact of the noise produced by its activity. The processing by-products are reused within the organization itself for the same function they were originally used for, in view of the principle of "remanufacturing" (R6).

The examination of Organization VC's sustainability report has shown that "*it welcomed the introduction of sharing platforms, as well as a reduction in plastic packaging*" and "*it reuses parts of products that it uses in other production processes and minimizes waste*". It is clear that Organization VC responds to the principle of "rethink" (R1) through the tool that allows it to contribute to the sharing economy. Sharing platforms operate through the internet, mobile applications, and social networks for the improvement of services and effective communication (Rao, 2007). The organization's policy is to move from a linear production cycle to a CE in which value maximization and resource efficiency are achieved to minimize (R2) single-use plastics. The goal is to reuse (R3), i.e., maximize recyclable (R8) packaging within the same organization and accelerate the development of bio-based biodegradable packaging. Often the parts of some products are reused in internal production processes, as per the "remanufacture" (R6) principle.

The sustainability report of the Organization L states that "the innovation of thought comprises the fundamental willingness to get involved in the various stages of corporate life; [...] all our factories are equipped with a water purification system, which allows us to minimize the impact on the surrounding ecosystem; our production plants are equipped with a complex system of heat recovery to make processing systems more efficient; there is also a whey concentrator, which collects the whey coming from the processing of all our plants,

optimizing the recovery process and significantly reducing the environmental impact due to transport; our packaging is from renewable sources". Thus, Organization L responds to the rethink (R1) principle through collaborative models of the sharing economy and sharing as a closed-loop activity that can help the community and the organization itself for the enhancement of underutilized assets. The company chooses materials and products with 100% recyclable (R8) packaging with a glass recycling rate of 85%; it uses 100% plant-based inks, produced using energy created from photovoltaic systems; it recycles and reduces (R2) the quantity of products sent to landfill and effectuates CO2. The company reuses its production waste in internal production, thereby responding to the "reuse" (R3) principle.

Finally, the analysis of Organization GM's sustainability report shows that "*important investments have been made to improve logistics and transport as well as contain the plant's environmental impact;* [...] *in addition to transport by ship, it is focusing on rail transport:* 50% still remains by road, but its goal is to reduce it" and "it has built a biomass power *plant, which is able to cover 70% of its energy needs and it has reduced its packaging by* 100%". Thus, the Organization GM responds to the "rethink" (R1) principle with a greater awareness of its stakeholders, who are aware that resources are limited and that the current linear production model is unsustainable. Substituting and acquiring goods through sharing is an innovative concept that also favours the reduction (R2) of consumption. The company also uses 75% recycled (R8) and completely recyclable materials, and promotes CO2 savings by increasing the use of hybrid cars or public transport. Their packaging is 100% reused (R3).

From the analysis of the above-mentioned results, it is clear that the publishing of sustainability reports (which are not mandatory for companies), in addition to achieving the benefits brought about by CSR, also favours the adoption of CE practices (Schallmo *et al.*, 2017). In greater detail, R1, R2 and R8 dimensions are present in all the companies considered. R3 is instead present in all of them but one. Finally, R6 and R7 are present in very

few of them, while the remaining dimensions for circularity are not evidenced at all in the sustainability reports.

--- Insert Figure 2 about here ---

5. Discussion

The concept of CE is linked to an innovative path towards sustainability that pursues not only the economic aims of the organization but also the environmental and social ones. Indeed, CE brings several benefits to all the stakeholders involved by aiding the environment and the society on one hand (Pomponi and Moncaster, 2017) and operations and production flows on the other (Parida *et al.*, 2019; Ranta *et al.*, 2018a), resulting to be an environmental innovation (Fernandez de Arroyabe *et al.*, 2021; Dias Angelo *et al.*, 2012). Therefore, knowing how to encourage the adoption of CE is increasingly critical for all the stakeholders (Del Vecchio *et al.*, 2022). The other managerial approach aiming at sustainable development on which we focused in this study is CSR. Indeed, recent studies have theoretically affirmed an interconnection between CSR and CE, and call for additional theoretical and empirical research in this direction (Daú *et al.*, 2019). The need to investigate the integration of these two frameworks for sustainability is also supported by the recent synergies that have been showed to arise between CSR and Research and Development (R&D) investments (Fu *et al.*, 2020).

Thus, this study fills in this cue by advancing the understanding of the synergistic application of two above-mentioned phenomena by empirically showing how CSR may favour the emergence of CE approaches in the Agri-food sector. In greater detail, not every dimension of CE is benefited and the adoption of the CE practice considered, i.e., 9Rs (Kirchherr *et al.*, 2017), is not progressive among companies but rather dispersed. In other words, CSR favors the emergence of CE, but in an unstructured way. These outcomes are in line with the theoretical studies that draw attention to the aforementioned interconnection, and we are

indeed providing empirical quantification of such relationship.

The contributions of this study are manifold, impacting theory, practice and policymaking, as detailed below in the following subsections.

5.1 Theoretical contributions

CSR is an established business management and governance model based on satisfying stakeholders' expectations. From the perspective of CSR, if a company does not pay particular attention to all interested parties, it will be doomed to fail. In this study, we have evidenced that such theoretical grounding can be extended to the CE phenomenon, which takes into account all the stakeholders involved and aims to benefit the whole society. We highlight how stakeholders' theory is a theoretical lens common to CSR and CE approaches, and can therefore be effectively used to understand how the two phenomena are correlated. None of the previous studies have exploited this theoretical grounding so far, which is indeed crucial to fully grasp the interconnections of the two phenomena. Based on this, we posit how CSR may favour CE, and by adopting an exploratory, qualitative research design we provide evidence of this. While it has been recently shown that there can be synergies between common efforts in CSR and R&D (Fu et al., 2020) due to common expertise and know-how, we have here evidenced how CSR and CE can be synergistically implemented. Thus, knowledge management of CE, towards which research has increasingly focused (Zucchella and Previtali, 2019; Govindan and Hasanagic, 2018; Zhang et al., 2021; Gomes et al., 2021; Vendrell-Herrero, 2021), might start from the CSR activities already undergone by the company. Moreover, we have evidenced that the positive impact of CSR on CE adoption is not structured, as might be thought. The benefits in terms of circularity do not follow the linear approach of the 9R model, evidencing that although CSR already benefits CE, further adjustments by managers and policymakers should be put in place to make CE benefit in a linear way. A further contribution of the study is the way we assessed the transition towards circularity, i.e., with the 9R model that was recently proposed by Kirchherr (2017), providing further evidence of the applicability of this model.

5.2 Contributions for practice

Considering the benefits that CE can bring to all the stakeholders and the whole society, managerial interest is increasingly embracing this new approach. However, the shift from linear economy towards CE can imply several costs for organizations. In light of this transition, the outcomes of this research can be extremely important for a managerial audience as they highlight how the efforts undertaken aimed at CSR can be also utilized for the transition towards circularity. In fact, the customer's propensity to purchase sustainable materials and products also leads the entrepreneur to refine the management practice of circularity models. Our study has empirically evidenced that CSR can be a useful lever to encourage the adoption of CE and that it is better for companies to start with CSR first to thus be already on the road to reach structured CE models. Therefore, companies that have already implemented a CSR approach can realize they are ready for the transition towards CE, since, on average, half of the approach should be already in place. Moreover, companies that have not yet started the transition towards either CSR or CE may be motivated to do so contemporaneously, because the costs for enhancing CSR will also help embrace circularity. We are confident that such results can be extremely useful to spread the concept of CE - as well as to further adopt CSR - among organizations. This is extremely relevant in light of the recent evidence that some companies have cut their CSR efforts (Fu et al., 2020).

5.3 Implications for policymaking

Policymakers are looking for ways to tackle the pressing grand challenges that our society is facing, and CE is an emerging strategy towards this end (e.g., Circular Economy Package in EU (De Schutter *et al.*, 2020). To that end, this study provides evidence that favouring the

emergence of CSR has the secondary effect of benefiting the transition towards CE. As companies may decide to shift their spending to the development of new products (Fu *et al.*, 2020), especially in the Covid-19 era where resources are scarce, showing to companies that efforts in CSR be also benefit the transition towards CE and consequently satisfying all the stakeholders, may be beneficial for the joint diffusion of CSR and CE. Thanks to this, it might be possible to benefit all three pillars of sustainability, i.e., economic, social and environmental, both through CSR and CE. Thus, more thorough knowledge of the determinants of the interconnection of CSR towards the CE could allow policymakers to more effectively communication and make directed interventions.

6. Conclusions

Companies should increasingly comply with the stakeholders' requests and favour sustainable growth, considering economic, political and social spheres. Towards that aim, CSR and CE can play crucial roles, especially if jointly implemented. The self-regulatory model of CSR is an older and already structured model, which may constitute a catalyst for the evolution and implementation of the more recent model of CE.

Towards this end, in the present research we have theoretically and empirically evidenced how CSR can promote CE approaches. In so doing, in addition to advancing the scholarly knowledge about these phenomena, we also provide practical implication aimed at favouring the implementation of CSR and CE and thus benefit all the stakeholders involved.

This study is not exempt from limitations, which nonetheless leave space for various future developments. First, while our study evidenced that CSR favours the adoption of CE practices, but not a complete transition towards circularity, future research may analyse whether companies that undergo CSR are more oriented towards CE than those who did not. In addition, future studies can analyse how fast and easy the complete adoption of circularity is for companies that have already developed some CE practice thanks to CSR. Such

outcomes will further strengthen the relevance and benefits of CSR for companies. Moreover, as this study has been conducted in the Agri-food sector in the Italian context, which is a relevant case study due to its technological advancements and the great attention towards all the stakeholders, future research can focus on different businesses and geographical locations to further validate the outcomes and examine whether differences exist related to other sectors and locations. Furthermore, while this study has been conducted on eight SMEs active in the agri-food, which is a sample sufficient for reaching theoretical saturation for this study and in line with recent qualitative research conducted in the context of Agri-food SMEs (O'Connor, and Kelly, 2017), future studies may consider enlarging the number of organizations considered to further validate the outcomes of this research. Finally, while in this research we have focused on SMEs, future research may consider other type of firms, e.g. listed firms, to evaluate differences as compared to the sector analysed in the current research.

References

Acciarini, C., Brunetta, F. and Boccardelli, P. (2020), "Cognitive biases and decision-making strategies in times of change: a systematic literature review", *Management Decision*, Vol. 59 No. 3, pp. 638-652. doi:10.1108/MD-07-2019-1006

Annosi, M.C., Brunetta, F., Monti, A. and Nat, F. (2019), "Is the trend your friend? An analysis of technology 4.0 investment decisions in agricultural SMEs", *Computers in Industry*, Vol. 109, pp. 59-71. doi:10.1016/j.compind.2019.04.003

Appio, F.P., Frattini, F., Petruzzelli, A.M. and Neirotti, P. (2021), "Digital Transformation and Innovation Management: A Synthesis of Existing Research and an Agenda for Future Studies", *Journal of Product Innovation Management*, Vol. 38 No. 1, pp. 4-20. doi:10.1111/jpim.12562

Ardito, L., Cerchione, R., Del Vecchio, P. and Raguseo, E. (2019), "Big data in smart tourism: challenges, issues and opportunities", *Current Issues in Tourism*, Vol. 22 No. 15, pp. 1805-1809. doi:10.1080/13683500.2019.1612860

Aung, M.M. and Chang, Y.S. (2014), "Traceability in a food supply chain: Safety and quality perspectives", *Food Control*, Vol. 39, pp. 172-184. doi:10.1016/j.foodcont.2013.11.007

Baden-Fuller, C. and Teece, D.J. (2020), "Market sensing, dynamic capability, and competitive dynamics", *Industrial Marketing Management*, Vol. 89, pp. 105-106. doi:10.1016/j.indmarman.2019.11.008

Barrena-Martínez, J., López-Fernández, M., Márquez-Moreno, C. and Romero-Fernández, P.M. (2015), "Corporate Social Responsibility in the Process of Attracting College Graduates", *Corporate Social Responsibility and Environmental Management*, Vol. 22 No. 6, pp. 408-423. doi:10.1002/csr.1355

Barreiro-Gen, M. and Lozano, R. (2020) "How circular is the circular economy? Analysing the implementation of circular economy in organisations", *Bus Strat Env*, Vol. 29, pp. 3484-3494. doi:10.1002/bse.25903494BARREIRO-GENANDLOZANO

Baumgartner, R.J. (2014), "Managing Corporate Sustainability and CSR: A Conceptual Framework Combining Values, Strategies and Instruments Contributing to Sustainable Development", *Corporate Social Responsibility and Environmental Management*, Vol. 21 No. 5, pp. 258-271. doi:10.1002/csr.1336

Berrone, P., Fosfuri, A., Gelabert, L. and Gomez-Mejia, L.R. (2013), "Necessity as the mother of "green" inventions: Institutional pressures and environmental innovations", *Strategic Management Journal*, Vol. 34, pp. 891-909. doi:10.1002/smj.2041

Bigerna, S., Micheli, S. and Polinori, P. (2020), "New generation acceptability towards durability and reparability of products: Circular economy in the era of the 4th industrial revolution", *Technological Forecasting and Social Change*, Vol. 165, 120558. doi:10.1016/j.techfore.2020.120558

Blomsma, F. (2018), "Collective 'action recipes' in a circular economy - On waste and resource management frameworks and their role in collective change", *Journal of Cleaner Production*, Vol. 199, pp. 969-982. doi:10.1016/j.jclepro.2018.07.145

Bolzani, D., Carli, G., Fini, R. and Sobrero, M. (2015). "Promoting Entrepreneurship in the Agri-food Industry: Policy Insights from a Pan-European Public-Private Consortium", *Industry and Innovation*, Vol. 22 No. 8, pp. 1366-2716. doi:10.1080/13662716.2015.1113860 Brenes, E.R., Ciravegna, L. and Acuña, J. (2020). "Differentiation strategies in agribusiness - A configurational approach", *Journal of Business Research*, Vol. 119, pp. 522-539. doi:10.1016/j.jbusres.2020.07.048

Brewster, C., Roussaki, I., Kalatzis, N., Doolin, K. and Ellis, K. (2017), "IoT in Agriculture: Designing a Europe-Wide Large-Scale Pilot", *IEEE Communications Magazine*, Vol. 55 No. 9, pp. 26-33. doi:10.1109/MCOM.2017.1600528

Burger, M., Stavropoulos, S., Ramkumar, S., Dufourmont, J. and van Oort, F. (2019), "The heterogeneous skill-base of circular economy employment", *Research Policy*, Vol. 48 No. 1, pp. 248-261. doi:10.1016/j.respol.2018.08.015

Cainelli, G., D'Amato, A. and Mazzanti, M. (2020), "Resource efficient eco-innovations for a circular economy: Evidence from EU firms", *Research Policy*, Vol. 49 No. 1, 103827. doi:10.1016/j.respol.2019.103827

Cappa, F., Rosso, F., Giustiniano, L. and Porfiri, M. (2020), "Nudging and Citizen Science: The Effectiveness of Feedback in Energy-Demand Management". *Journal of Environmental Management*, Vol. 269, 110759. doi:10.1016/j.jenvman.2020.110759

Carroll, A.B. (1991), "The Pyramid of Corporate Social Responsibility: Toward the Moral Management of Organizational Stakeholders", *Business Horizons*, Vol. 34 No. 4, pp. 39-48. doi:10.1016/0007-6813(91)90005-G

Cassar, L. and Meier, S. (2018), "Stop Talking About How CSR Helps Your Bottom Line", *Harvard Business Review*, available at: https://hbr.org/2018/01/stop-talking-about-how-csr-helps-your-bottom-line (18 February 2022)

Centobelli, P., Cerchione, R., Chiaroni, D., Del Vecchio, P. and Urbinati, A. (2020), "Designing business models in circular economy: A systematic literature review and research agenda", *Business Strategy and the Environmental*, Vol. 29 No. 4, pp. 1734-1749. doi:10.1002/bse.2466

Ciliberto, C., Szopik-Depczyńska, K., Tarczyńska-Łuniewska, M., Ruggieri, A. and Ioppolo, G. (2021), "Enabling the Circular Economy transition: a sustainable lean manufacturing recipe for Industry 4.0", *Business Strategy and the Environment*, Vol. 30 No. 7, pp. 3255-3272. https://doi.org/10.1002/bse.2801

Ciravegna, L. and Brenes, E.R. (2016), "Learning to become a high reliability organization in the food retail business". *Journal of Business Research*, Vol. 69 No. 10, pp. 4499-4506. doi:10.1016/j.jbusres.2016.03.015

Ciulli, F. and Kolk, A. (2019), "Incumbents and Business model innovation for the sharing economy: Implications for sustainability", *Journal of Cleaner Production*, Vol. 214, pp. 995-1010. doi:10.1016/j.jclepro.2018.12.295

Colbert, A., Yee, N. and George, G. (2016), "The digital workforce and the workplace of the future", *Academy of Management Journal*, Vol. 59 No. 3, pp. 731-739.

Commission of the European Communities (2001), "Green Paper on Integrated Product Policy", available at: https://ec.europa.eu/environment/ipp/2001developments.htm (accessed 9 June 2022)

Cordeiro, J.J. and Tewari, M. (2015), "Firm characteristics, industry context, and investor reactions to environmental CSR: A stakeholder theory approach", *Journal of Business Ethics*, Vol. 130 No. 4, pp. 833-849. doi:10.1007/s10551-014-2115-x

Corsini, F., Laurenti, R., Meinherz, F., Appio, F.P., and Mora, L. (2019). "The advent of practice theories in research on sustainable consumption: Past, current and future directions of the field", *Sustainability*, Vol. 11 No. 2, p. 341. doi:10.3390/su11020341

D'Amato, D., Droste, N., Allen, B., Kettunen, M., Lähtinen, K., Korhonen, J. and Toppinen, A. (2017), "Green, circular, bio economy: A comparative analysis of sustainability avenues", *Journal of Cleaner Production*, Vol. 168, pp. 716-734. doi:10.1016/j.jclepro.2017.09.053

Daú, G., Scavarda, A., Scavarda, L.F. and Portugal, V.J.T. (2019), "The healthcare sustainable supply chain 4.0: The circular economy transition conceptual framework with the corporate social responsibility mirror", *Sustainability*, Vol. 11 No. 12, p. 3259. doi:10.3390/su11123259

Del Vecchio, P., Urbinati, A. and Kirchherr, J. (2022), "Enablers of Managerial Practices for Circular Business Model Design: An Empirical Investigation of an Agro-Energy Company in a Rural Area", *IEEE Transactions on Engineering Management*. doi: 10.1109/TEM.2021.3138327.

Del Vecchio, P., Di Minin, A., Petruzzelli, A.M., Panniello, U. and Pirri, S. (2018), "Big data for open innovation in SMEs and large corporations: Trends, opportunities, and challenges", *Creativity and Innovation Management*, Vol. 27, pp. 6-22. doi:10.1111/caim.12224

De Schutter, O., Jacobs, N., and Clément, C. (2020), "A 'Common Food Policy' for Europe: How governance reforms can spark a shift to healthy diets and sustainable food systems", *Food Policy*, Vol. 96, 101849. doi:10.1016/j.foodpol.2020.101849

Dias Angelo, F., Jose Chiappetta Jabbour, C. and Vasconcellos Galina, S. (2012), "Environmental innovation: in search of a meaning", *World Journal of Entrepreneurship*,

Management and Sustainable Development, Vol. 8 No. 2-3, pp. 113-121. doi:10.1108/20425961211247734

Dmytriyev, S.D., Freeman, R.E. and Hörisch, J. (2021), "The Relationship between Stakeholder Theory and Corporate Social Responsibility: Differences, Similarities, and Implications for Social Issues in Management", *Journal of Management Studies*, Vol. 58 No. 6. doi:10.1111/joms.12684

Elia, G., Polimeno, G., Solazzo, G. and Passiante, G. (2019), "A multi-dimension framework for value creation through big data", *Industrial Marketing Management*, Vol. 15, pp. 589-598. doi:10.1016/j.indmarman.2019.08.004

Ellen MacArthur Foundation (2015), "Why the circular economy matters? Delivering the Circular Economy: A Toolkit for Policymakers", available at: https://www.ellenmacarthurfoundation.org/assets/downloads/publications/EllenMacArt (accessed 9 June 2022)

Ellen-MacArthur Foundation (2013), "Toward the circular economy", available at: https://www.ellenmacarthurfoundation.org/assets/downloads/publications/Ellen-MacArthur-Foundation-Towards-the-Circular-Economy-vol.1.pdf (accessed 9 June 2022)

Esken, B., Franco-García, M.L. and Fisscher, O.A.M. (2018), "CSR perception as a signpost for circular economy", *Management Research Review*, Vol. 41 No. 5, pp. 586-604. doi:10.1108/MRR-02-2018-0054

European Commission (2021), "Annual report on European SMEs 2020/2021", available at: file:///C:/Users/Donato/Downloads/SME%20Annual%20Report%20-%202021%20(1).pdf (accessed 9 June 2022)

European Commission (2020), "Circular economy action plan", available at: https://doi.org/10.2775/855540 (accessed 9 June 2022)

European Commission (2019), "The EU Fish Market - 2019", available at: https://www.eumofa.eu/documents/20178/314856/EN_The+EU+fish+market_2019.pdf (accessed 9 June 2022)

European Commission (2015), "An EU action plan for the circular economy", available at: https://eur-lex.europa.eu/resource.html?uri=cellar:8a8ef5e8-99a0-11e5-b3b7-

01aa75ed71a1.0012.02/DOC 1&format=PDF (accessed 9 June 2022)

Fait, M., Scorrano, P., Mastroleo, G., Cillo, V. and Scuotto, V. (2019), "A novel view on knowledge sharing in the agri-food sector", *Journal of Knowledge Management*, Vol. 23 No. 5, pp. 953-974. https://doi.org/10.1108/JKM-09-2018-0572

FEEM (2020), "The New Circular Economy Action Plan", available at: https://www.feem.it/m/publications pages/brief09-2020.pdf (accessed 9 June 2022).

Fehrer, J.A. and Wieland, H. (2021), "A systemic logic for circular business models", *Journal* of Business Research, Vol. 125, pp. 609-620. doi:10.1016/j.jbusres.2020.02.010

Ferasso, M., Beliaeva, T., Kraus, S., Clauss, T., Ribeiro-Soriano, D. (2020), "Circular economy business models: The state of research and avenues ahead", *Bus Strat Env*, Vol. 29, pp. 3006-3024. https://doi.org/10.1002/bse.2554

Fernandez de Arroyabe, J.C., Arranz, N., Schumann, M. and Arroyabe, M.F. (2021), "The development of CE business models in firms: The role of circular economy capabilities", *Technovation* Vol. 106, 102292. doi:10.1016/j.technovation.2021.102292

Fleming, D. (2001), "Narrative leadership: Using the power of stories", *Strategy & Leadership*, Vol. 29 No. 4. doi:10.1108/sl.2001.26129dab.002

Flick, U., von Kardorff, E. and Steinke, I. (2004), *A Companion to Qualitative Research*, Sage Publications Ltd, Thousand Oaks, CA.

Food and Agriculture Organization of the United Nations (2020), "The state of the food and Agriculture", available at: http://www.fao.org/publications/sofa/en/ (accessed 9 June 2022)

2
3
1
5
6
7
8
9
10
10
11
12
13
14
15
16
10
17
18
19
20
21
22
-∠ 22
23
24
25
26
27
28
20
29
30
31
32
33
34
25
35
36
37
38
39
40
/1
+1 40
42
43
44
45
46
17
-T/ /0
40
49
50
51
52
53
55
54
55
56
57
58
59
~ ~

60

Franco, S., Caroli, M.G., Cappa, F. and Del Chiappa, G. (2020), "Are you good enough? CSR, quality management and corporate financial performance in the hospitality industry", *International Journal of Hospitality Management*, Vol. 88, 102395. doi:10.1016/j.ijhm.2019.102395

Frederick, W.C. (1960), "The Growing Concern over Business Responsibility", *California Management Review*, Vol. 2 No. 4, pp. 54-61. doi:10.2307/41165405

Freeman, R.E. and Dmytriyev, S. (2017), "Corporate social responsibility and stakeholder theory: Learning from each other. Symphonya", *Emerging Issues in Management*, Vol. 1, pp. 7-15. doi:10.4468/2017.1.02freeman.dmytriyev

Freeman, E.R. and Evan W. M. (1990), "Corporate governance: A stakeholder interpretation", *The Journal of Behavioral Economics*, Vol. 19 No. 4: pp. 337-359. doi:10.1016/0090-5720(90)90022-Y

Friedman, M. (2017), "The social responsibility of business is to increase its profits", Zimmerli, W.C., Holzinger, M. and Richter, K. (Eds.), *Corporate Social Responsibility*, pp. 173-178. doi:10.1007/978-3-540-70818-6 14

Fu, L., Boehe, D. and Orlitzky, M. (2020), "Are R&D-Intensive firms also corporate social responsibility specialists? A multicountry study", *Research Policy*, Vol. 49 No. 8, 104082. doi:10.1016/j.respol.2020.104082

Gangi, F., Mustilli, M. and Varrone, N. (2019), "The impact of corporate social responsibility (CSR) knowledge on corporate financial performance: evidence from the European banking industry", *Journal of Knowledge Management*, Vol. 23 No. 1, pp. 110-134. doi:10.1108/JKM-04-2018-0267

Gangi, F., Mustilli, M., Varrone, N. and Daniele, L.M. (2018), "Corporate Social Responsibility and Banks' Financial Performance", *International Business Research*, Vol. 11 No. 10, pp. 42-58. doi:10.5539/ibr.v11n10p42

Geissdoerfer, M., Savaget, P., Bocken, N.M.P. and Hultink, E.J. (2017), "The Circular Economy - A new sustainability paradigm?" *Journal of Cleaner Production*, Vol. 143, pp. 757-768. doi:10.1016/j.jclepro.2016.12.048

Geissinger, A., Laurell, C., Öberg, C., and Sandström, C. (2019), "How sustainable is the sharing economy? On the sustainability connotations of sharing economy platforms", *Journal of Cleaner Production*, Vol. 206, pp. 419-429. doi:10.1016/j.jclepro.2018.09.196

Ghasemzadeh, P., Rezayat Sorkhabadi, S.M., Kebriaeezadeh, A., Nazari, J.A., Farzaneh, M. and Mehralian, G. (2021), "How does organizational learning contribute to corporate social responsibility and innovation performance? The dynamic capability view", *Journal of Knowledge Management*. doi:10.1108/JKM-01-2021-0069

Ghisellini, P., Cialani, C. and Ulgiati, S. (2016), "A review on circular economy: The expected transition to a balanced interplay of environmental and economic systems", *Journal of Cleaner Production*, Vol. 114, pp. 11-32. doi:10.1016/j.jclepro.2015.09.007

Gomes, L.A.d.V., de Faria, A.M., Borini, F.M., Flechas Chaparro, X.A., dos Santos, M.G. and Gurgel Amaral, G.S. (2021), "Dispersed knowledge management in ecosystems", *Journal of Knowledge Management*, Vol. 25 No. 4, pp. 796-825. https://doi.org/10.1108/JKM-03-2020-0239

Govindan, K. and Hasanagic, M. (2018), "A systematic review on drivers, barriers, and practices towards circular economy: a supply chain perspective". *International Journal of Production Research*, Vol. 56 No. (1-2), pp. 278-311. doi:10.1080/00207543.2017.1402141 Grover, V. and Davenport, T.H. (2001), "General Perspectives on Knowledge Management: Fostering a Research Agenda", *Journal of Management Information Systems*, Vol. 18 No. 1, pp. 5-21. https://doi.org/10.1080/07421222.2001.11045672

Hanley, A. and Semrau, F.O. (2022), "Stepping up to the mark? Firms' export activity and environmental innovation in 14 European countries", *Industry and Innovation*, Vol. 29 No. 1, pp. 1366-2716. doi:10.1080/13662716.2021.2021865.

Halloran, A., Clement, J., Kornum, N., Bucatariu, C. and Magid, J. (2014), "Addressing food waste reduction in Denmark", *Food Policy*, Vol. 49 No. 1, pp. 294-301. doi:10.1016/j.foodpol.2014.09.005

Hansmann, R., Mieg, H.A. and Frischknecht, P. (2012), "Principal sustainability components: Empirical analysis of synergies between the three pillars of sustainability", *International Journal of Sustainable Development & World Ecology*, Vol. 19, pp. 451-459. doi:10.1080/13504509.2012.696220

Heslin, P.A. and Ochoa, J.D. (2008), "Understanding and developing strategic corporate social responsibility", *Organizational Dynamics*, Vol. 37 No. 2, pp. 125-144. doi:10.1016/j.orgdyn.2008.02.002

Heyes, G., Sharmina, M., Mendoza, J.M.F., Gallego-Schmid, A. and Azapagic, A. (2018), "Developing and implementing circular economy business models in service-oriented technology companies", *Journal of Cleaner Production*, Vol. 177, pp. 621-632. doi:10.1016/j.jclepro.2017.12.168

Hussainey, K. and Salama, A. (2010), "The importance of corporate environmental reputation to investors", *Journal of Applied Accounting Research*, Vol. 11, pp. 229-241. doi:10.1108/09675421011088152

Kamath, R. (2018), "Food Traceability on Blockchain: Walmart's Pork and Mango Pilots with IBM", *The Journal of the British Blockchain Association*, Vol. 1 No. 1, pp. 47-53. doi:10.31585/jbba-1-1-(10)2018

Kamilaris, A., Fonts, A. and Prenafeta-Boldú, F.X. (2019), "The rise of blockchain technology in agriculture and food supply chains", *Trends in Food Science and Technology*, Vol. 91, pp. 640-652. doi:10.1016/j.tifs.2019.07.034

Kane, G. C., Palmer, D., Phillips, A. N., Kiron, D. and Buckley, N. (2015), "Strategy, not technology, drives digital transformation", *MIT Sloan Management Review and Deloitte University Press*, Vol. 14, pp. 1-25.

Kikuchi, Y. and Kanematsu, Y. (2019), "Life cycle assessment", Kozai, T., Niu, G. and Takagaki, M. (Eds.), *Plant Factory - An Indoor Vertical Farming System for Efficient Quality Food Production* (Second Edition), pp. 383-395. doi:10.1016/B978-0-12-816691-8.00027-3 Kirchherr, J., Reike, D. and Hekkert, M. (2017), "Conceptualizing the circular economy: An analysis of 114 definitions", *Resources, Conservation and Recycling*, Vol. 127, pp. 221-232. doi:10.1016/j.resconrec.2017.09.005

Kong, D., Shi, L. and Yang, Z. (2019), "Product recalls, corporate social responsibility, and firm value: Evidence from the Chinese food industry", *Food Policy*, Vol. 83, pp. 60-69. doi:10.1016/j.foodpol.2018.11.005

Kong, D. (2012), "Does corporate social responsibility matter in the food industry? Evidence from a nature experiment in China", *Food Policy*, Vol. 37 No. 3, pp. 323-334. doi:10.1016/j.foodpol.2012.03.003

Korhonen, J., Honkasalo, A. and Seppälä, J. (2018), "Circular Economy: The Concept and its Limitations", *Ecological Economics*, Vol. 143, pp. 37-46. doi:10.1016/j.ecolecon.2017.06.041 Kristoffersen, E., Blomsma, F., Mikalef, P. and Li, J. (2020), "The smart circular economy: A digital-enabled circular strategies framework for manufacturing companies", *Journal of Business Research*, Vol. 120, pp. 241-261. doi:10.1016/j.jbusres.2020.07.044

Lawrence, J. (2013), "Cradle to cradle", *TCE The Chemical Engineer*. doi:10.9774/gleaf.978-1-907643-44-6_38

2
z
4
4
5
6
7
,
ð
9
10
11
11
12
13
14
15
10
16
17
18
10
20
20
21
22
22
2J 24
24
25
26
27
27
28
29
30
21
31
32
33
34
25
35
36
37
20
20
39
40
41
12
+2
43
44
45
16
40
47
48
49
50
50
51
52
52 53
52 53
52 53 54
52 53 54 55
52 53 54 55 56
52 53 54 55 56
52 53 54 55 56 57
52 53 54 55 56 57 58

60

Leandro, A. and Paixao, S. (2018), "Corporate Social Responsibility and Circular Economy: Two ways, same destinations? An outlook on both concepts and cases from Portugal", paper presented at the Congrèsavni R, 7 November, Lille, Portugal

Lewandowski, M. (2016), "Designing the business models for circular economy-towards the conceptual framework", *Sustainability*, Vol. 8 No. 1, p. 43. doi:10.3390/su8010043

Linstead, S., Maréchal, G. and Griffin, R.W. (2014), "Theorizing and researching the dark side of organization", *Organization Studies*, Vol. 35 No. 2, pp. 165-188. doi:10.1177/0170840613515402

Liu, D.Y., Chen, S.W. and Chou, T.C. (2011), "Resource fit in digital transformation: Lessons learned from the CBC Bank global e-banking project", *Management Decision*, Vol. 49 No. 10, pp. 1728-1742.

Loiseau, E., Saikku, L., Antikainen, R., Droste, N., Hansjürgens, B., Pitkänen, K. and Thomsen, M. (2016), "Green economy and related concepts: An overview", *Journal of Cleaner Production*, Vol. 139, pp. 361-371. doi:10.1016/j.jclepro.2016.08.024

Ltifi, M. and Hichri, A. (2022), "The effects of corporate governance on the customer's recommendations: a study of the banking sector at the time of COVID-19", *Journal of Knowledge Management*, Vol. 26 No. 1, pp. 165-191. doi:10.1108/JKM-06-2020-0471

Lubin, D.A and Esty, D.C. (2012), "Bridging the Sustainability Gap", *MIT Sloan Management Review*, Vol. 55, pp. 18-21.

Lucchini, A. and Moisello, A. (2017), "CSR Disclosure, Visibility and Media Pressure International Evidence from the Apparel and Textile Industry", *European Journal of Economics, Finance and Administrative Sciences*, Vol. 93, pp. 5-28.

Manninen, K., Koskela, S., Antikainen, R., Bocken, N., Dahlbo, H. and Aminoff, A. (2018), "Do circular economy business models capture intended environmental value propositions?" *Journal of Cleaner Production*, Vol. 171, pp. 413-422. doi:10.1016/j.jclepro.2017.10.003

Maon, F., Swaen, V. and De Roeck, K. (2021), "Corporate branding and corporate social responsibility: Toward a multi-stakeholder interpretive perspective", *Journal of Business Research*, Vol. 126, pp. 64-77. doi:10.1016/j.jbusres.2020.12.057

Marrucci, L., Daddi, T. and Iraldo, F. (2021), "The circular economy, environmental performance and environmental management systems: the role of absorptive capacity", *Journal of Knowledge Management*. doi:10.1108/JKM-06-2021-0437

McGuire, J.W. (1969), "The Changing Nature of Business Responsibilities", *The Journal of Risk and Insurance*, Vol. 36 No. 1, pp. 55-61. doi:10.2307/251140

Modgil, S., Gupta, S. and Bharat, B. (2021), "Big data-enabled large-scale group decision making for circular economy: An emerging market context", *Technological Forecasting and Social Change*, Vol. 166, 120607. doi:10.1016/j.techfore.2021.120607

Morseletto, P. (2020), "Targets for a circular economy", *Resources, Conservation and Recycling*, Vol. 153, 104553. doi:10.1016/j.resconrec.2019.104553

Mylona, K., Maragkoudakis, P., Miko, L., Bock, A.K., Wollgast, J., Caldeira, S. and Ulberth, F. (2018), "Viewpoint: Future of food safety and nutrition - Seeking win-wins, coping with trade-offs", *Food Policy*, Vol. 74, pp. 143-146. doi:10.1016/j.foodpol.2017.12.002

Nasir, V.A. and Karakaya, F. (2014), "Consumer segments in organic foods market", *Journal of Consumer Marketing*, Vol. 31 No. 4, pp. 263-277. doi:10.1108/JCM-01-2014-0845

Newell, S. (2015), "Managing knowledge and managing knowledge work: what we know and what the future holds", *Journal of Information Technology*, Vol. 30 No. 1, pp. 1-17. doi:10.1057/jit.2014.12

O'Connor, C. and Kelly, S. (2017), "Facilitating knowledge management through filtered big data: SME competitiveness in an agri-food sector", *Journal of Knowledge Management*, Vol. 21 No. 1, pp. 156-179. https://doi.org/10.1108/JKM-08-2016-0357

Pan, S.Y., Gao, M., Kim, H., Shah, K.J., Pei, S. L. and Chiang, P.C. (2018), "Advances and challenges in sustainable tourism toward a green economy", *Science of the Total Environment*, Vol. 635, pp. 452-469. doi:10.1016/j.scitotenv.2018.04.134
Parida, V., Burström, T., Visnjic, I. and Wincent, J. (2019), "Orchestrating industrial ecosystem in circular economy: A two-stage transformation model for large manufacturing

companies", Journalof Business Research, Vol. 101, pp. 715-725. doi:10.1016/j.jbusres.2019.01.006

Patwa, N., Sivarajah, U., Seetharaman, A., Sarkar, S., Maiti, K. and Hingorani, K. (2021), "Towards a circular economy: An emerging economies context", *Journal of Business Research*, Vol. 122, pp. 725-735. doi:10.1016/j.jbusres.2020.05.015

Pellegrini, G., Annosi, M.C., Contò, F. and Fiore, M. (2020), "What are the conflicting tensions in an Italian cooperative and how do members manage them? Business goals', integrated management, and reduction of waste within a fruit and vegetables supply chain", *Sustainability*, Vol. 12 No. 7, 3050. doi:10.3390/su12073050

Perrini, F., Russo, A. and Tencati, A. (2007), "CSR strategies of SMEs and large firms. Evidence from Italy", *Journal of Business Ethics*, Vol. 74, pp. 285-300. doi:10.1007/s10551-006-9235-x

Pieroni, M.P.P., McAloone, T.C. and Pigosso, D.C.A. (2019), "Business model innovation for circular economy and sustainability: A review of approaches", *Journal of Cleaner Production*, Vol. 215, pp. 198-216. doi:10.1016/j.jclepro.2019.01.036

Pomponi, F., and Moncaster, A. (2017), "Circular economy for the built environment: A research framework", *Journal of Cleaner Production*, Vol. 143, pp. 710-718. doi:10.1016/j.jclepro.2016.12.055

Porter, M.E. and Kramer, M.R. (2006), "The Link Between Competitive Advantage and Corporate Social Responsibility", *Harvard Business Review*, available at: www.hbr.orgwww.fsg-impact.org (accessed 9 June 2022)

Postolache, A.G. and Troaca, V.A. (2018), "Green economy", *Quality - Access to Success*, Vol. 19, pp. 423-427. doi:10.14512/oew.v29i3.1300

Radu, C. and Smaili, N. (2021), "Corporate performance patterns of Canadian listed firms: Balancing financial and corporate social responsibility outcomes", *Business Strategy and the Environment*, Vol. 30 No. 7, pp. 3344-3359. https://doi.org/10.1002/bse.2806

Rainville, A. (2021), "Stimulating a more Circular Economy through Public Procurement: Roles and dynamics of intermediation", *Research Policy*, Vol. 50 No. 4, 104193. doi:10.1016/j.respol.2020.104193

Ranta, V., Aarikka-Stenroos, L., Ritala, P. and Mäkinen, S.J. (2018), "Exploring institutional drivers and barriers of the circular economy: A cross-regional comparison of China, the US, and Europe", *Resources, Conservation and Recycling*, Vol. 135, pp. 70-82. doi:10.1016/j.resconrec.2017.08.017

Rao, N.H. (2007), "A framework for implementing information and communication technologies in agricultural development in India", *Technological Forecasting and Social Change*, Vol. 74 No. 4, pp. 491-518. doi:10.1016/j.techfore.2006.02.002

Román, S., Sánchez-Siles, L.M. and Siegrist, M. (2017), "The importance of food naturalness for consumers: Results of a systematic review", *Trends in Food Science and Technology*, Vol. 67, pp. 44-57. doi:10.1016/j.tifs.2017.06.010

Russo, A. and Perrini, F. (2010), "Investigating stakeholder theory and social capital: CSR in large firms and SMEs", *Journal of Business Ethics*, Vol. 91, No. 2, pp. 207-221. doi:10.1007/s10551-009-0079-z

2
3
4
5
6
7
, 0
0
9
10
11
12
13
14
15
16
17
10
10
19
20
21
22
23
24
25
26
20
27
28
29
30
31
32
33
34
25
22
36
37
38
39
40
41
42
43
ΔΛ
44
45
46
47
48
49
50
51
52
52 52
55 7
54
55
56
57
58
50

60

Saidani, M., Yannou, B., Leroy, Y., Cluzel, F. and Kendall, A. (2019), "A taxonomy of circular economy indicators", *Journal of Cleaner Production*, Vol. 207, pp. 542-559. doi:10.1016/J.JCLEPRO.2018.10.014

Sakshi, Shashi, Cerchione, R. and Bansal, H. (2020), "Measuring the impact of sustainability policy and practices in tourism and hospitality industry", *Business Strategy and the Environmental*, Vol. 29, pp. 1109-1126. doi:10.1002/bse.2420

Salvioni, D. and Almici, A. (2020), "Circular Economy and Stakeholder Engagement Strategy. Symphonya", *Emerging Issues in Management*, Vol. 1, pp. 26-44. doi:10.4468/2020.1.03salvioni.almici

Santos, M. (2011), "CSR in SMEs: Strategies, practices, motivations and obstacles", *Social Responsibility Journal*, Vol. 7 No. 3, pp. 490-508. doi:10.1108/1747111111154581

Schallmo, D., Williams, C. A. and Boardman, L. (2017), "Digital transformation of business models - bestpractice, enablers, and roadmap", *International Journal of Innovation Management*, Vol. 21 No. 08, 1740014. doi:10.1142/S136391961740014X

Seles, B. M. R. P., Mascarenhas, J., Lopes de Sousa Jabbour, A.B. and Trevisan, A.H. (2022), "Smoothing the circular economy transition: The role of resources and capabilities enablers", *Business Strategy and the Environment*, pp. 1-24. https://doi.org/10.1002/bse.2985

Sharma, G., Beveridge, Alim J. and Haigh, N. (2018), "A configural framework of practice change for B corporations", *Journal of Business Venturing*, Vol. 33 No. 2, pp. 207-224. doi:10.1016/j.jbusvent.2017.12.008

Shepherd, M., Turner, J. A., Small, B. and Wheeler, D. (2018), "Priorities for science to overcome hurdles thwarting the full promise of the 'digital agriculture' revolution", *Journal of the Science of Food and Agriculture*, Vol. 100 No. 14, pp. 5083-5092. doi:10.1002/jsfa.9346

Siggelkow, N. (2007), "Persuasion With Case Studies", *The Academy of Management Journal*, Vol. 50 No. 1, pp. 20-24. doi:10.5465/amj.2007.24160882

Song, C., Gardner, K.H., Klein, S.J.W., Souza, S.P. and Mo, W. (2018), "Cradle-to-grave greenhouse gas emissions from dams in the United States of America", *Renewable and Sustainable Energy Reviews*, Vol. 90, pp. 945-956. doi:10.1016/j.rser.2018.04.014

Song, C.C.S. (1972), "The Limits to Growth", *JAWRA Journal of the American Water Resources Association*, Vol. 8 No. 4, p. 837. doi:10.1111/j.1752-1688.1972.tb05230.x

Spalding, M.D., Ruffo, S., Lacambra, C., Meliane, I., Hale, L.Z., Shepard, C.C. and Beck, M.W. (2014), "The role of ecosystems in coastal protection: Adapting to climate change and coastal hazards", *Ocean and Coastal Management*, Vol. 90, pp. 50-57. doi:10.1016/j.ocecoaman.2013.09.007

Steurer, R. (2010), "The role of governments in corporate social responsibility: Characterising public policies on CSR in Europe", *Policy Sciences*, Vol. 43, pp. 49-72. doi:10.1007/s11077-009-9084-4

Stoyanova, T. (2019), "CSR Strategies Applied in Termes of Circular Economy", *Economic Alternatives*, Vol. 2, pp. 263-274.

Suchek, N., Fernandes, C. I., Kraus, S., Filser, M. and Sjögrén, H. (2021) "Innovation and the circular economy: A systematic literature review", *Business Strategy and the Environment*, Vol. 30 No. 8, pp. 3686-3702. https://doi.org/10.1002/bse.2834

Surasak, T., Wattanavichean, N., Preuksakarn, C. and Huang, S.C.H. (2019), "Thai agriculture products traceability system using blockchain and Internet of Things", *International Journal of Advanced Computer Science and Applications*, Vol. 10 No. 9, pp. 578-583. doi:10.14569/ijacsa.2019.0100976

Theodoulidis, B., Diaz, D., Crotto, F., and Rancati, E. (2017), "Exploring corporate social responsibility and financial performance through stakeholder theory in the tourism industries", *Tourism Management*, Vol. 62, pp. 173-188. doi:10.1016/j.tourman.2017.03.018 Tian, F. (2016), "An agri-food supply chain traceability system for China based on RFID & blockchain technology", paper presented at the13th International Conference on Service Systems and Service Management (ICSSSM), Kunming, 2016, 1-6. doi:10.1109/ICSSSM.2016.7538424

Tukker, A. (2015), "Product services for a resource-efficient and circular economy - A review", *Journal of Cleaner Production*, Vol. 97, pp. 76-91. doi:10.1016/j.jclepro.2013.11.049

Urbinati, A., Chiaroni, D. and Chiesa, V. (2017), "Towards a New Taxonomy of Circular Economy Business Models", *Journal of Cleaner Production*, Vol. 168, pp. 487-98. doi:10.1016/j.jclepro.2017.09.047

Vázquez-Burguete, J.L., Sahelices-Pinto, C., and Lanero-Carrizo, A. (2017), "Corporate social responsibility and consumer behavior in the cosmetics sector: a study in the Spanish context", *International Review on Public and Nonprofit Marketing*, Vol. 14 No. 3, pp. 375-390. doi:10.1007/s12208-017-0178-y

Velenturf, A.P.M., Jensen, P.D., Purnell, P., Jopson, J. and Ebner, N. (2019), "A Call to Integrate Economic, Social and Environmental Motives into Guidance for Business Support for the Transition to a Circular Economy", *Administrative Sciences*, Vol. 9 No. 4, p. 92. doi:10.3390/admsci9040092

Vendrell-Herrero, F., Gomes, E., Opazo-Basaez, M. and Bustinza, O.F. (2021), "Knowledge acquisition throughout the lifecycle: product and industry learning frameworks", *Journal of Knowledge Management*, Vol. ahead-of-print No. ahead-of-print. https://doi.org/10.1108/JKM-05-2021-0387

Venkatraman, V. (2017), *The digital matrix: new rules for business transformation through technology*, Wonderwell, Los Angeles, CA.

Waheed, A. And Zhang, Q. (2020), "Effect of CSR and ethical practices on sustainable competitive performance: A case of emerging markets from stakeholder theory perspective", *Journal of Business Ethics*. doi:10.1007/s10551-020-04679-y

Warner, K.S. and Wäger, M. (2019), "Building dynamic capabilities for digital transformation: An ongoing process of strategic renewal", *Long Range Planning*, Vol. 52 No. 3, pp. 326-349. doi:10.1016/j.lrp.2018.12.001

Werther, W.B. and Chandler, D. (2005), "Strategic corporate social responsibility as global brand insurance", *Business Horizons*, Vol. 48 No. 4, pp. 317-324. doi:10.1016/j.bushor.2004.11.009

WHO - Regional Committee for Europe (2014). "European food and nutrition action plan2015-2020",Availableat:

https://www.euro.who.int/__data/assets/pdf_file/0003/294474/European-Food-Nutrition-Action-Plan-20152020-en.pdf (accessed 9 June 2022).

Yin, R.K. (2009). *Case Study Research: design and Methods, 4th ed.*, Sage Publications, Thousand Oaks, CA.

Zhang, B., Comite, U., Yucel, A.G., Liu, X., Khan, M.A., Husain, S., Sial, M.S., Popp, J. and Oláh, J. (2021), "Unleashing the Importance of TQM and Knowledge Management for Organizational Sustainability in the Age of Circular Economy", *Sustainability*, Vol. 13 No. 20,11514. https://doi.org/10.3390/su132011514

Zucchella, A and Previtali, P. (2019), "Circular business models for sustainable development: A "waste is food" restorative ecosystem", *Bus Strat Env.*, Vol. 28, pp. 274-285. https://doi.org/10.1002/bse.2216

1	
2 3	Annendix
4	
5 6	Insert Table AI about here
7	
8	
9 10	
11	
12 12	
14	
15	
16 17	
18	
19 20	
20 21	
22	
23 24	
25	
26	
27 28	
29	
30 31	
32	
33	
34 35	
36	
37 38	
39	
40	
41 42	
43	
44 45	
46	
47	
49	
50	
51 52	
53	
54	
56	
57	
58	

- 58 59
- 60

Figure 1. The 9R model used in this study to assess the inclination towards circularity (source: Kirchherr et al. 2017).

Figure 2. Results of the analysis about firms' orientation towards CE, referring to the 9R model (Kirchherr et al. 2017) (source: authors' elaboration).

											20	
Circul	ur T	:T TURE	0 USE Strategies	Organization F	Organization GC	Organization DMA	Organization S	Organization VDOSMTA	Organization VC	Organization L	Organization CM	
economy		DUC	REF									
	SMARTERPRO USE AND MANUF		SMARTERPRO E AND MANUI	RETHINK	x	x	x	x	x	x	x	x
		R2 REDUCE	x	x	x	x	x	x	x	x		
		лст	R3 REUSE	x		x	x	x	x	x	x	
ng circularity			REPAIR									
		SPAN OF PROI ITS PARTS	R5 REFURBISH									
Increas		EXTENDLIFE AND	R6 REMANUFAC TURE	x			x	x	x			
			R7 REPURPOSE			x	x					
	TIONOF	VTIONOF LS	R8 RECYCLE	x	x	x	x	x	x	x	x	
Linea	r	USEFUL APPLICA MATERIAI	R9 RECOVER									

Page 46 of 46

Appendix

Table AI. Comparison of the main characteristics of the sampled firms (source: authors' elaboration).

Companies	Headquarters	Production sector	Consumer Products	Reference markets		
Organization F	Verrucchio (Rimini)	Cereal sector Production and sale of semi-finished products for pastry and ice cream	Spreads and liqueurs Decorations for pastry	70% by the national market and the remaining 30% by the world		
Organization GC	Maierato (Vibo Valentia)	Produces and sells gut, roe, freshly processed tuna	Mackerel fillets, anchovy fillets	Products are marketed on the national territory, while 7% of the total turnover i absorbed by foreign markets (Canada, Austria, France, Switzerland, Australia Slovenia, United States, Lithuania, Grea Britain, and South Africa).		
Organization DMA Flumeri (Avellino)		Durum wheat flours	Conventional dry pasta, egg pasta, vitamin zed pasta	The markets to which it is addressed at mainly foreign (United States, United Kingdom, Europe, Japan, Arab countrie where about60% of the production is exported. The rest is marketed in Italy		
Organization S	Ascoli Piceno (Marche)	Dairy sector	Milk and dairy products	Its reference markets are central and northern Italy, while 1% of its turnover exported to France and Austria		
Organization VDOSMTA Creazzo (Vicenza)		Wine and spirits sectorSells its products (wines) mainly in Italy to the large-scale retail trade		Its target market is mainly Italy		
Organization VC	Isola della Scala (Verona)	Wine and spirits sector	The first company in Italy for the creation of braille labels.	Its target market is mainly Italy		
Organization L Busche Cesiomaggiore (Belluno)		Dairy sector	Milk and dairy products	Market with products of excellence recognized and appreciated by many Italian and international consumers		
Organization Rovigo GM (Veneto)		Cereal sector	It produces soft and durum wheat flour Kamut, organic flour	Sells products in the national and international market		